10 17:00:48 Gvr 2014
100

Thu Apr
Year: 14 Doy:
st ueben

Cbserver:
WEATHER COMVENT:
Clear, wind south 5
___end___
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Thu Apr
Thu Apr
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GENERAL

Thu Apr
Fri Apr
Fri Apr
Fri Apr
Fri Apr
WEATHER

end

**PSPT PROBLEM COMVENT BY st ueben**
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35:
29:
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45:
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OBSERVATORY

COMVENT:

st ueben:

14
46
31
31
26
42
43

22
12
21
21
57

st ueben:

Thu Apr

nph, tenp 41F.

GMI' 2014:

GMI' 2014
GVl 2014
GV 2014
GMI 2014
GMI 2014
GMI' 2014

PSPT Start

KCOR Start Synoptic Patrol

10 17:00:56 GVIr 2014

Pat r ol

KCOR End Patr ol

KCOR Start Calibration script:
KCOR End Calibration Script
KCOR Start Synoptic Patrol
KCOR Start Synoptic Patrol

COVMENT BY st ueben:
Adj ust ed SGS of fsets.
end

GV 2014:
GMI 2014:
GVl 2014:
GVl 2014:

PSPT Abort
PSPT Start
PSPT Abort
PSPT Start

Thu Apr

Pat r ol
Pat r ol
Pat r ol
Pat r ol

GMI' 2014 KCOR End Patr ol

Fri

Apr 11 00:47:51 GV 2014
O ographi ¢ clouds are beginning to degrade i nmages,

Started to investigate focus problem |
where the focus appears to begin to stick and then conmanded noves with the

"fcs goto xxxx" command on the pc. | could not get the focus to mi sbehave, it
al ways went to the commanded position.

end

Fri Apr 11 01:02:17 GMVI 2014: PSPT Abort Patrol
11 01:02:22 GMr 2014: PSPT Abort Patrol
WEATHER COMVENT:
I n cl ouds,
___end___
ColP COWMENT BY stueben: Fri
St eve downl oaded and installed the new drivers and we can now talk to the

Fri Apr

meadow ark control |l ers.

___end___

st ueben:
i nstrunents idled.

Fri

Apr

Apr 11 01:05:07 GMI 2014

Fri

c:\kcor\m so-calibration22deg. i ni

10 22:05:47 GMr 2014

i dl ed.

Apr 11 00:49:13 GMIr 2014

11 01:05:51 GVr 2014

noved the tel escope to the position

There are a few mnor tweeks to be done tonorrow but
conp is basically working again with its new conputer.



